Author’s response to reviews

Title: Incidence and risk factors of spinal epidural hemorrhage after spine surgery: A crosssectional retrospective analysis of a national database

Authors:

Ji Hyun Park (pharmerpark@korea.ac.kr)
Sunny Park (psunny0708@korea.ac.kr)
Soo An Choi (sachoi@korea.ac.kr)

Version: 2 Date: 21 Apr 2020

Author’s response to reviews:

April 22nd

Dear Editor,

We’ve corrected Materials and methods to "Methods" in the abstract, and "Data and Methods" to "Methods" in the main body.

Thank you so much!

Sincerely,

Soo An Choi, Ph.D., Prof. College of Pharmacy, Korea University Tel.: +82 44-860-1626; Fax: +82 44-860-1606; E-mail: sachoi@korea.ac.kr